Author's response to reviews

Title: Reliability of movement control tests in the cervical spine

Authors:

Maja Patroncini (maja.patroncini@ksw.ch)
Susanne Hannig (susannehannig@freenet.de)
Andre Meichtry (mcdr@zhaw.ch)
Hannu Luomajoki (luom@zhaw.ch)

Version: 5 Date: 13 November 2014

Author's response to reviews:

Dear Editor,

Thank you for regarding our paper to be published in your Journal. We have now revised our manuscript. The only concern of yours was the exact name of the ethic commission which approved the study. We have now added this information saying:

The study was performed in accordance with the Declaration of Helsinki and approval from the local ethics commission (ethical commission of canton Zürich, Switzerland) was received.

with best regards, Hannu Luomajoki